Author's response to reviews

**Title:** Relationship between bone mineral density, its associated physiological factors, and tooth loss in postmenopausal Korean women

**Authors:**

Chang-Suk Kim (molar5103@hanmail.net)
Eun-Kyong Kim (jinha01@naver.com)
Kyeong-Soo Lee (drkslee@ynu.ac.kr)
Hee-Kyung Lee (lhk3731@yu.ac.kr)
Youn-Hee Choi (cyh1001@knu.ac.kr)
Tae-Yoon Hwang (luke@ynu.ac.kr)
Jun Sung Moon (mjs@yu.ac.kr)

**Version:** 7  
**Date:** 26 June 2015

**Author's response to reviews:** see over
Dear editor of BMC women’s health,

The authors of this study sincerely appreciate your constructive suggestion. We tried to do our best to modify and improve this manuscript. What we have done further is described as below;

Editorial Requirement:

*Copyediting: We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Answer: We requested copyediting of our manuscript to Edanz which BMC recommended. We are very happy to have chance to improve the style of written English of manuscript.